Author's response to reviews

Title: The New Zealand Asthma and Allergy Cohort Study (NZA2CS): Assembly, Demographics and Investigations

Authors:

Michael J Epton (michael.epton@chmeds.ac.nz)
George I Town (ian.town@canterbury.ac.nz)
Tristram Ingham (Tristram_Ingham@wnmeds.ac.nz)
Kristin Wickens (kwickens@wnmeds.ac.nz)
David Fishwick (David.Fishwick@hsl.gov.uk)
Julian Crane (crane@wnmeds.ac.nz)
The New Zealand Asthma and Allergy Cohort Study Group (nza2cs) (crrg@chmeds.ac.nz)

Version: 4 Date: 27 February 2007

Author's response to reviews:

The Editors

BMC Public Health

Re: The New Zealand Asthma and Allergy Cohort Study (NZA2CS): Assembly, Demographics and Investigations

Thank you for your email of 20th February about the above manuscript. We have made the proofing and formatting changes suggested by the Editorial Production Team. We hope they are acceptable. The Author web-page would not accept the Author Group abbreviation submission in capitals (NZA2CS), and requested an email address for the author group - we have supplied the email address of the Canterbury Respiratory Research Group, which we hope suffices.

We hope the changes prove acceptable and look forward to the publication of the manuscript.

Yours sincerely

Dr Michael Epton

On behalf of the Authors